

TO EVALUATE MUSCULOSKELETAL MODELS OF CARE

Curtin University





Health

South Western Sydney Local Health District







SUPPORTING ORGANISATIONS

The following organisations publicly support this evaluation framework.





MACQUARIE UNIVERSITY



MEDICAL UNIVERSITY OF VIENNA



NATIONAL RHEUMATOID ARTHRITIS SOCIETY (UK)



North East Quality Observatory Service

NORTH EAST QUALITY OBSERVATORY SERVICE (NEQOS) NHS (UK)



PAINAUSTRALIA



PAN AMERICAN LEAGUE OF ASSOCIATIONS FOR RHEUMATOLOGY (PANLAR)



ROMANIAN SOCIETY OF REHABILITATION MEDICINE



OSAKA HEALTH SCIENCE UNIVERSITY



PAINLESS



RHEUMATOLOGY HEALTH PROFESSIONALS ASSOCIATION



Chiropractors

ROYAL COLLEGE OF CHIROPRACTORS UK



OSTEOARTHRITIS RESEARCH SOCIETY INTERNATIONAL (OARSI)



PAKISTAN SOCIETY FOR THE REHABILITATION OF THE DISABLED



ROMANIAN OCCUPATIONAL THERAPY ASSOCIATION



SOUTHERN DENMARK UNIVERSITY



SYDNEY NORTH HEALTH NETWORK



THE EHLERS-DANLOS SOCIETY



UNITED STATES BONE AND JOINT INITIATIVE



UNIVERSITY OF MELBOURNE



WORLD CONFEDERATION FOR PHYSICAL THERAPY



World Federation of Occupational Therapists

WORLD FEDERATION OF OCCUPATIONAL THERAPISTS



ZIMBABWE ASSOCIATION OF OCCUPATIONAL THERAPY

Please direct correspondence to:

Associate Professor Andrew Briggs Email: A.Briggs@curtin.edu.au

Suggested citation for this report

Briggs AM, Jordan JE, Jennings M, Speerin R, Chua J, Bragge P, Slater H (2016): A Framework to Evaluate Musculoskeletal Models of Care. Cornwall, UK: Global Alliance for Musculoskeletal Health of the Bone and Joint Decade.

CONTENTS

Foreword

5

Executive summary and using this report

Th	ne Framework at a glance	7
	What is a Model of Care?	7
	Who uses Models of Care?	7
	Purpose of this project and the Framework	7
	What is the Framework designed to do and why should I use it?	8
	Development of the Framework	8
	How to use the Framework	8
Ho	ow to use this document and the Framework	9
	The document as a whole	9
	The Framework in Part 3	9
	avigating the Framework for the evaluation musculoskeletal Models of Care	11
	checklist of essential items for evaluating odels of Care	12

2

Setting the scene

Background Information	
Context	14
Models of Care	14
The Framework	15
Intended audience	16
Our approach to developing the Framework	16

3

The Framework

READINESS STREAM

1	Stru	cture and components of the MoC document	21
	1A	A clear outline	21
	1B	A data-driven case for change	22
	1C	Define the target population/priority groups	23
	1D	Cost-effectiveness data	24
2	Eng	agement and consultation	25
	2A	Important stakeholders	25
	2B	What to ask and explore	26
	2C	Seeking endorsement	27
	2D	Identifying and supporting local champions	27
3		moting best practice by describing <i>what</i> care <i>how</i> to deliver it	28
	3A	Align to contemporary standards	28
	3B	Identify required behaviour changes	28
	3 C	Utilise different service delivery modes	29
	3D	Specify communication and referral pathways	29
4	Con	sumer centric	30
	4A	Practical, user-friendly recommendations	30
	4B	Partnership-based service delivery and funding	30

INITIATING IMPLEMENTATION STREAM

5 Op	timising implementation and evaluation success	32
5A	Assess system readiness	32
5B	Linking to local resources	32
5C	Identifying likely workforce requirements	33
5D	Building a comprehensive implementation plan	34
5E	Formative evaluation of MoC components	35
5F	Establishing a User Reference Group	35

3

The Framework

SUCCESS STREAM

6	Con	tinuous improvement processes	37
	6A	Pragmatic evaluations over time	37
	6B	Quality assurance and troubleshooting mechanisms	38
	6C	Data collection processes for key performance indicators	39
	6D	Promoting research priorities	39
7	Key	performance indicators	40
	7A	Consumer relevant outcomes	40
	7B	Service delivery partnerships and pathways	41
	7C	Cost-effectiveness	42
	7D	Stakeholder behaviour changes	43
8	Eng	agement and participation	44
	8A	Awareness and knowledge of the MoC	44
	8B	Reach to target population	45
	8C	Satisfaction with processes and programs	45
9	Upt	ake and integration	46
	9A	Adaptability across settings and responsiveness	46
	9B	Innovative changes to service resourcing	47
	9C	The MoC becomes routine business	47
	9D	The MoC is utilised as a resource	48
	9E	The new MoC replaces the old MoC	48

Putting the Framework into practice

Scenario 1: Using the Framework to judge the readiness of a Model of Care for osteoarthritis	51
Scenario 2: Using the Framework to assess preparedness for implementation or prepare for implementation	56
Scenario 3: Using the Framework to judge the success of a Model of Care for osteoporosis	60

5 Su Defi Ack Ref

Supporting information

Definitions of terms	65
Acknowledgements	67
References	69

The burden of disease associated with chronic non-communicable diseases (NCDs), particularly musculoskeletal conditions, is now clear. Indeed, data from the most recent analyses of the Global Burden of Disease study unequivocally reinforce this issue.

Urgent and coordinated global action is required to address the rising burden of disease associated with these conditions to ensure health services can meet the current and future needs of health consumers. Supporting low and middle-income nations to develop appropriate responses now is essential.

Models of Care represent one approach to respond to the burden of NCDs. Models of Care outline the principles of best practice management for specific conditions, thus providing guidance for 'what works' and 'how to implement it'.

Although many nations are developing Models of Care to address NCDs, there remains inconsistency in the approach to their development and evaluation, making comparisons between them difficult. Further, achieving sustainable implementation is challenging. For these reasons, development of an internationally-informed framework to evaluate the 'readiness' of Models of Care for implementation and their 'success' after implementation is of international importance. The Global Alliance for Musculoskeletal Health of the Bone and Joint Decade is pleased to be a partner on this project that aimed to develop such a framework. While the focus of the Framework has been on musculoskeletal health, the end products have relevance to Models of Care for NCDs generally.

As a global community, our call to action is to now use the Framework to support and optimise our development, implementation and evaluation endeavours to improve the lives of people who are at risk of, or live with, chronic NCDs.

Tony hoof.

Professor Anthony D. Woolf Chair

Global Alliance for Musculoskeletal Health of the Bone and Joint Decade





EXECUTIVE SUMMARY AND USING THIS REPORT Models of Care are increasingly viewed as an effective strategy to improve health service planning and delivery for non-communicable diseases. Despite the increased attention towards Models of Care, a universal framework to evaluate a Model's readiness for implementation and success after implementation is lacking. This Framework addresses these important gaps.

THE FRAMEWORK AT A GLANCE

What is a Model of Care?

A Model of Care (**MoC**) is a principle-based guide that describes best practice care for particular health conditions or populations. The focus is on person-centred care and consideration of applicability in local settings. A MoC is not an operational plan for a health service or a clinical practice guideline.

Who uses Models of Care?

MoCs have cross-sector and multi-stakeholder relevance. Policy makers, health administrators and managers, service delivery organisations, clinicians, researchers, funders, advocacy organisations and consumers use MoCs to inform best practice planning and delivery of health services.

Purpose of this project and the Framework

To develop a comprehensive evaluation framework to assess the *readiness* for implementation and *success* after implementation of musculoskeletal MoCs. The Framework provides principle-based guidance on evaluating these important areas. Particular emphasis is placed on ensuring the Framework is applicable across a diverse range of environments and contexts.

What is the Framework designed to do and why should I use it?

The Framework is designed to help individuals and organisations tasked with the planning, implementation or evaluation of MoCs.

Specifically, the Framework can be used to:

- Develop a clear and concise MoC document that is acceptable to local stakeholders.
- Judge whether a MoC is ready for implementation → *Readiness Stream*.
- Guide the initial implementation process → Initiating Implementation Stream.
- Consider performance measures that are likely to indicate the MoC is successful → Success Stream.

Part 4 of this report, "Putting the Framework into practice" provides practical examples of how the Framework could be used in practice.

Development of the Framework

The Framework was developed using a four-phase approach, drawing on the knowledge and experiences of 93 international experts across 30 countries.

- Phase 1: Identification of the important concepts that underpin 'readiness' and 'success' of MoCs, based on in-depth interviews with Australian experts.
- **Phase 2:** Assessment of these concepts and their further development with an international panel of experts using an eDelphi method.
- **Phase 3:** Translation of the concepts into a usable and meaningful Framework for end users using a Knowledge-to-Action approach.
- **Phase 4:** Testing of the accuracy and acceptability of the Framework with the international expert panel.

How to use the Framework

The Framework has three streams:

- i. Readiness.
- ii. Initiating implementation.
- iii. Success.

Each stream has a number of domains and each domain has a number of themes. Each domain and theme is numbered to allow easy navigation across the Framework (Figure 1). Use the map on page 11 to identify relevant parts for your work.

Themes marked with a gold star have been identified as essential to a particular stream (see essential checklist on page 12). Other themes should be viewed as important, but not necessarily essential in all settings.



Figure 1: Example of Framework layout

The document as a whole

The document is divided into five parts:

- Part 1 is the executive summary.
- Part 2 provides the background to the project.
- Part 3 contains the Framework.
- Part 4 provides scenarios of how the Framework could be applied in practice.
- Part 5 contains supporting information definitions, acknowledgements and references.

The Framework in Part 3

The Framework contains three STREAMS:

Stream 1. *Readiness* (blue section): this stream outlines what should be included in a regional or national MoC, how it should be presented and the process of development. This stream is relevant to developers at a national or regional level.

Stream 2. *Initiating Implementation* (orange section): this stream describes how to approach implementation after a MoC has been developed.

It provides guidance on what to consider for optimising implementation success and how to develop an implementation plan. This stream is relevant to those tasked with implementation of a MoC, usually at a local or regional level.

Stream 3. *Success* (green section): this stream considers how to approach evaluation, including both formative evaluation and impact evaluation that includes consumer and system-relevant outcomes. This stream is relevant to those tasked with monitoring the outcomes of a MoC, usually at a local or regional level.

Important notes for interpreting the Framework (Part 3):

- Within each stream are a number of DOMAINS.
- Within each domain are a number of THEMES.
- Essential themes are indicated by a gold star.
- A number of PRINCIPLES underpin each theme.

Figure 2 below shows how the Framework is structured using this hierarchy.



Detailed principles underpinning each theme

Figure 2: Structure of the Framework illustrating a stream, domain, theme and principles. Here, the Readiness stream is used as an example.

Additional section for Success stream

The *Success stream* contains additional information on performance indicators/methods/data. This additional information recommends the "how to" with respect to undertaking evaluation activities (Figure 3).



Additional information on performance indicators/methods/data

Figure 3: Schematic of the Success stream illustrating the additional section related to performance indicators/methods/data.

NAVIGATING THE FRAMEWORK FOR THE EVALUATION OF MUSCULOSKELETAL MODELS OF CARE

READINESS STREAM

Structure and components of a MoC document

- 1A A clear outline
- 1B A data-driven case for change
- **1C** Define the target population/priority groups
- 1D Cost-effectiveness data

) Engagement and consultation

- **2A** Important stakeholders
- 2B What to ask and explore
- 2C Seeking endorsement
- **2D** Identifying and supporting local champions

Promoting best practice care by describing *what* care and *how* to deliver

- **3A** Align to contemporary standards
- 3B Identify required behaviour changes
- 3C Utilise different service delivery modes
- **3D** Specify communication and referral pathways

Consumer centric

3

4

- 4A Practical, user-friendly recommendations
- **4B** Partnership-based service delivery and funding

INITIATING IMPLEMENTATION STREAM

- 5 Optimising implementation and evaluation success
 - 5A Assessing system readiness
 - **5B** Linking to local resources
 - **5C** Identifying likely workforce requirements
 - **5D** Building a comprehensive implementation plan
 - **5E** Formative evaluation of MoC components
 - **5F** Establishing a multidisciplinary User Reference Group

SUCCESS STREAM

6

7

8

9

)	Cor	ntinuous improvement process
	6A	Pragmatic evaluations over time
	6B	Quality assurance and troubleshooting mechanisms
	6C	Data collection for key performance indicators
	6D	Promoting research priorities
)	Key	v performance indicators
	7A	Consumer relevant outcomes
	7B	Service delivery partnerships and pathways
	7C	Cost-effectiveness
	7D	Stakeholder behaviour change
)	Eng	agement and participation
	8A	Awareness and knowledge of the MoC
	8B	Reach to target population
	8C	Satisfaction with processes and programs
)	Upt	ake and integration
	9A	Adaption across settings
	9B	Innovative changes to service resourcing
	9C	The MoC becomes routine business
	9D	The MoC is utilised as a resource

- **9E** The new MoC replaces the previous MoC
- Figure 4: Orientation map for the Framework illustrating the 3 streams (3 colour bands), domains within the streams (blocks) and themes in the domains.

A CHECKLIST OF ESSENTIAL ITEMS FOR EVALUATING MODELS OF CARE

The checklist below is a quick reference tool that contains only the essential evaluation areas, as determined by the expert panel that informed the development of the Framework. The checklist should be used in conjunction with the full Framework (Part 3 of this report), rather than a stand-alone resource.

READINESS STREAM	ESSENTIAL ★
1A The MoC document should provide a clear outline of aims, processes and outcome	es.
1B The MoC document should outline a well-developed and objective 'case for chang argument based on local, regional or national circumstances.	je'
1C The MoC should clearly define the target population and identify any specific prior	rity groups.
2A The MoC should be informed by meaningful engagement and consultation with a range of stakeholders.	broad
3A The MoC should align with standards of care for quality and safety and best practic specific musculoskeletal health conditions.	ce for
4A The MoC should be consumer-centred in all aspects and user-focused when description recommendations for implementation.	ibing

INITIATING IMPLEMENTATION STREAM

5D An implementation plan should be developed which includes guiding principles to inform the development of locally-relevant project or business plans to facilitate implementation of specific components of the MoC.

ESSENTIAL

SUC	CESS STREAM	ESSENTIAL 🛧
6A	A pragmatic evaluation has been undertaken at different time points, inclusive of outco and process evaluations.	mes
6 B	The MoC has ongoing quality assurance and troubleshooting processes.	
6C	Data collection processes have been established to measure pre-defined key performa indicators (KPIs).	ince
7 A	Over time, there is evidence of improved consumer experiences, access, health outcon and quality of life.	nes
7D	Once fully implemented, there is behaviour change amongst stakeholders, led initially by opinion leaders, aligned to the recommendations of the MoC.	
8 A	There is an awareness of the MoC amongst stakeholders and organisations (inclusive of consumers) in the long term.	
9A	The MoC has adaptability to be implemented in different contexts/environments/cultur and evolves over time.	es



SETTING THE SCENE



Context

The global burden associated with chronic, non-communicable diseases is enormous; particularly those conditions associated with morbidity, of which musculoskeletal conditions are ranked amongst the highest¹. For example, in the most recent Global Burden of Disease Study (GBD 2013), musculoskeletal conditions accounted for 23.3% of global Years Lived with Disability (**YLDs**) for non-communicable diseases (**NCDs**), second only to mental and substance use disorders (25.8% of YLDs)¹. Further, low back pain was identified as the leading condition associated with morbidity. Importantly, improving musculoskeletal health enables more effective management of other chronic health conditions².

Models of Care

The term "Model of Care" is not new. It has been used for many years to describe different aspects of health service delivery³. Other terms, such as 'care pathways' or 'standards of care' have also been used interchangeably with 'Models of Care'. While terminologies and scope may vary between nations, most of these general health service frameworks have a common purpose, which is to deliver effective and efficient consumer-centred healthcare.

In the context of this project we define a Model of Care as:

A Model of Care (**MoC**) is a person-centred and principle-based guide, usually presented as a document that describes:

- evidence-informed, best practice care for particular health conditions (in this case, musculoskeletal conditions)
- what care should be provided
- *how* it should be delivered at a regional or national level.

A MoC is not an operational plan for a health service or a clinical practice guideline. MoCs are usually implemented as health services at a local level.

MoCs are increasingly viewed as an effective strategy to improve health service planning and delivery for NCDs. MoCs are being developed across the globe, for a range of health conditions. In the context of the clinical focus of this project, musculoskeletal health, a broad range of MoCs have been developed for high income^{4,5,6} as well as middle- and low-income nations^{7,8}.

Generally, MoCs are developed in response to an identified population health need, for example, osteoarthritis care. They are designed to be nationally or regionally relevant, and therefore, contextually 'sit' within regional or national health systems.

MoCs are reflective of regional or national health policies, governance, funding, infrastructure and workforce characteristics and cultural sensitivities. When integrated into health systems in this manner, MoCs can serve as important platforms on which to drive reforms in health service planning and delivery.

Despite the increased attention directed towards the development, implementation and evaluation of MoCs,^{2,4,5,9} there remains no globally-informed framework to evaluate and guide these processes[‡]. This project developed an evaluation framework to address this gap. Specifically, two contemporary issues were considered:

1. Optimising development for sustainable implementation.

A range of system, organisational and provider factors can be a barrier to achieving sustainable implementation of MoCs¹⁰. There is merit, therefore, in optimising the 'readiness' of a MoC for implementation by considering what might be an ideal MoC development process and one, which will also enable sustainable implementation.

‡ At the time of this report, a Canadian tool had been developed in 2012⁹. However, this tool was developed specifically for Canada and limited in detail to a checklist approach.

2. Evaluating outcomes of Models of Care.

Evaluation of MoCs is critical as a step to inform and drive health reform and share evaluation experiences between countries. Identifying indicators of success and guiding how to measure these indicators is important for:

- a. local stakeholders such as governments
- b. sharing outcomes
- c. undertaking benchmarking internationally and across jurisdictions.

The Framework

The Framework is intended to complement both MoCs in development and those already implemented.

The Framework is designed to assist users in:

- judging whether a MoC is *ready* for implementation
- preparing for implementation
- planning evaluation and/or considering whether a MoC has been *successful* in implementation.

The Framework also supports the World Health Organisation's (**WHO**) approach to assessing nations' challenges and opportunities for health systems to improve outcomes related to NCDs¹¹. Whereas the WHO approach is based on a broad policy response to NCDs inclusive of cancer, cardiovascular disease, chronic obstructive pulmonary disease and diabetes, this Framework is specific to musculoskeletal MoCs.

Guiding Principles

The Framework is grounded on six guiding principles:

- The Framework is focused on improving consumer health outcomes and system-relevant outcomes for non-communicable diseases, particularly musculoskeletal conditions.
- While the focus of the Framework is on musculoskeletal health conditions, it is equally applicable to other non-communicable diseases that are best managed through a Chronic Care approach¹².
- **3.** The Framework is intentionally detailed to provide a 'best case scenario' or 'gold standard' approach to evaluating the readiness and success of MoCs. As such, it should be interpreted as a guide or resource with particular attention paid to components of the Framework that have been identified as 'essential'.
- **4.** The Framework is designed to be flexible, so it may be applied to MoCs in different settings and in developing and developed nations.
- **5.** The Framework is empirically-derived by a panel of international experts to reflect best evidence and practice and a 'real world' pragmatic approach to evaluation.
- 6. In considering its application to MoCs, the Framework is designed with the intention that it may articulate with regional or national health policy related to chronic health conditions and existing regional or national system performance frameworks.

Intended audience

The Framework is relevant to any person or individual who is tasked with the development, implementation or evaluation of MoCs. As such, the Framework is applicable to multiple stakeholder groups across different settings including:

- policy makers/governments
- service organisations
- clinical societies/associations/networks
- consumer advocacy groups and other non-government organisations
- researchers
- health funders/administrators.

While consumers, consumer organisations and clinicians have informed the Framework, it is not intended to be a resource for individual consumers and clinicians.

Our approach to developing the Framework

Consistent with Guiding Principle 5, the Framework was developed in a formal research context. Human Research Ethics Committee approval was granted to undertake the research by Curtin University, Western Australia. The research was conducted in partnership with tertiary institutions, government departments, a tertiary hospital and consultancy organisations in Australia, as well as with the Global Alliance for Musculoskeletal Health of the Bone and Joint Decade.

Our approach involved four phases, as outlined 1-4 below (Figure 5).



Figure 5: The four development phases informing the Framework.

Phase 1

Senior-level, multidisciplinary experts (policy makers, health service managers, insurers, advocacy organisations, clinicians, researchers and consumers) in musculoskeletal MoCs were sampled from three Australian states. Structured in-depth telephone interviews were conducted with 27 experts to explore:

- perceptions on the evaluation of MoCs
- concepts that should be included in an assessment of 'readiness' for implementation and 'success' after implementation.

Verbatim transcripts of the interviews were analysed to derive key themes relating to evaluation of readiness and success. These key themes and the associated detailed commentaries from the transcripts were used to develop an initial draft Framework. Further detail about this phase and its outcomes has been published¹³.

Phase 2

In collaboration with the Global Alliance for Musculoskeletal Health, an international panel of experts in musculoskeletal MoCs was established. The panel included the Australian experts from Phase 1. A total of 93 individuals across 30 countries participated, as displayed in Figure 6 below. Of the 30 countries represented, 63.3% were from high-income economies, 26.7% from middle-income economies and 10.0% from low-income economies, based on the 2016 World Bank classifications.

Using a standard eDelphi process, the panel iteratively provided feedback and scoring on the themes related to readiness and success that had been previously derived from Phase 1. Panellists also voted on which themes were 'essential' for the Framework. A theme was classified as 'essential' when \geq 80% of panellists nominated this category.



Figure 6: A global heat map showing locations of the expert Delphi panel. Darker shading reflects a greater proportion of panellists in a given location. Note the inclusion of developing and developed countries.

Phase 3

The multidisciplinary project team participated in a workshop to translate the data from Phases 1 and 2 into a meaningful Framework for end users. The methods used to design the workshop were based on a Knowledge-to-Action approach¹⁴, informed by evidence from the Australasian Cochrane Centre^{15,16}.

This involved development of a briefing document synthesizing key issues. Subsequently, a high-level, facilitated day-long discussion was undertaken in which information and questions for deliberation articulated in the briefing document were discussed to achieve the explicit aim of orienting the data into a usable Framework for end users^{17, 18}.

Phase 4

The same international panel from Phase 2 then scored and commented on the final version of the Framework. Key outcomes from the Delphi phase are summarised in the box below.

Key results from the Delphi panel

- Panel members strongly agreed with the draft Framework, with 'partly agree' or 'completely agree' scores ranging from 96.7-100% across the themes.
- 14 themes were classified as 'essential'.
- Panel members strongly agreed or agreed that the final Framework report was useful (98.8%), usable (95.1%), credible (100%) and appealing (93.9%).
- 96.3% strongly supported or supported the structure of the Framework as it was presented.
- 100%, 96.3% and 100% strongly supported or supported content within the readiness, initiating implementation and success streams, respectively.



THE FRAMEWORK



READINESS STREAM

This stream is relevant to developers at a national or regional level. This stream outlines:

- what should be included in a regional or national MoC
- *how* it should be presented and the process of development.



KEY DOMAINS

- 1. Structure and components of a MoC document
- 2. Engagement and consultation
- 3. Promoting best practice by describing what care and how to deliver it
- 4. Consumer centric

1. STRUCTURE AND COMPONENTS OF THE MOC DOCUMENT

A clear outline

The MoC document should provide a clear outline of aims, processes and outcomes.

1A

Principles:

- **1** The MoC document should communicate:
 - a clearly defined scope
 - aims and objectives
 - definitions
 - anticipated outcomes that are consumer-relevant as well as system-relevant and that facilitate measurement over time
 - a commentary about how the 'new' MoC replaces 'current care'

ESSENTIAL

- a commentary on the continuum of care being addressed by the model.
- 2 Each component of the MoC has clearly identified, consensus-based key performance indicator(s) (**KPI**) that are measurable in formative and impact evaluations over time.
- **3** The MoC should identify the extent to which it aligns with current health policy, strategy and frameworks.
- **4** The MoC clearly outlines its core components (i.e. the 'must have') to achieve the anticipated outcomes.

A data-driven case for change



The MoC document should outline a well-developed and objective 'case for change' argument based on local, regional or national circumstances.

1B

- 1 The 'case for change' should provide justification for a change to current service delivery and outline best practice, as informed by information available at the time, including:
 - consumers' needs
 - contemporary evidence for best practice
 - current local practice behaviours in the sector that are discordant with contemporary evidence
 - jurisdictional or national data reflecting burden of disease and system impacts.
- 2 There should be evidence that the current local service and provider environment is well understood, demonstrated by a comprehensive review of existing services, practice culture and referral behaviours. Aggregate information (informed by local audits and qualitative research with stakeholders) should include:
 - who provides care
 - where is care provided
 - how is care accessed
 - *what* are the blocks to care co-ordination
 - *where* are the care gaps/access blocks
 - *how* does the consumer typically move through the system now and *what* are their needs
 - *what* are the current governance arrangements around service delivery
 - what are the currently available funding streams.
- **3** Recommendations should not be based solely on evidence from randomised controlled trials (**RCTs**). The MoC should reflect a balance between academic literature and current practices in terms of what works and what is feasible to integrate in the current system.
- **4** Where the MoC refers to evidence for effective/ineffective interventions it should, where possible, make clinically-usable recommendations i.e. provide guidance on what interventions or behaviours *should* be provided, rather than focusing on what *not* to do.
- **5** The MoC should detail the processes used to identify and interpret evidence quality and relevance.
- Development of the MoC should include a horizon scan of new opportunities (infrastructure, technologies, therapies, policies) and MoCs in other healthcare systems to ensure it is contemporary.

Define the target population/priority groups

The MoC should clearly define the target population and identify any specific priority groups.

1C

- 1 The target population should be broadly defined, including specific information about health conditions within and outside the scope of the MoC (including comorbidities).
- **2** Justification for the target group should be informed by local data that demonstrate where services are needed the most.
- **3** The MoC should identify the factors that may affect reach to the target population e.g. access, cost, geography and knowledge of service through qualitative engagement with the target population.
- **4** The MoC should consider how the target population may vary across urban, regional and rural geographies as well as variance across indigenous and non-indigenous communities and communities of different socioeconomic statuses and ages.



Cost-effectiveness data

Where feasible, the cost-effectiveness of the proposed MoC should be outlined based on modelling, or existing evidence relevant to the local setting (e.g. data acquired in developed nations cannot be reliably translated to developing nations). Where reliable data are unavailable, the MoC should make recommendations about how to evaluate this in implementation.

1D

- Where possible, professional health economics advice should be sought in the development phase of the MoC and in planning cost-effectiveness evaluation in implementation trials.
- 2 The MoC should outline the likely local cost-effectiveness of the proposed model and identify where and how it is more cost-effective than current care, including cost to consumers. If reliable cost-effectiveness/cost savings data are not available, the MoC should provide recommendations on how to evaluate these factors as implementation commences e.g. through pilot studies.
- **3** Where cost savings are examined, cost shifting must be considered. For some, MoCs cost savings may not be expected.
- **4** The longer-term economic viability of the proposed MoC should be discussed.
- **5** The descriptors around cost-effectiveness and resource efficiencies should be based on system-relevant metrics that are meaningful to service administrators (e.g. cost per episode of care).

Important stakeholders

The MoC should be informed by meaningful engagement and consultation with a broad range of stakeholders (individuals and organisations) from inception to final development.

ZA

Principles:

- **1** The MoC should be informed by stakeholders across the care continuum and identified through thorough mapping of typical consumer pathways.
- 2 The MoC should outline a robust method of stakeholder identification, engagement and consultation across all development stages. At a minimum, stakeholders should include:
 - consumers/carers
 - subject matter experts or opinion leaders in musculoskeletal health
 - local clinical and administrative champions who are supported by their organisation to adopt a leadership position in the development, communication and implementation of the MoC
 - non-expert clinicians (e.g. generalists)
 - academic and non-academic clinicians
 - service administrators/managers
 - jurisdictional funders/policy makers
 - individuals who are likely to be tasked with implementation of a MoC in target settings
 - representative organisations (e.g. clinical organisations)
 - non-government sector (e.g. consumer advocacy organisations)
 - private service providers and insurers.
- **3** The nomination and selection processes related to expert working groups should be documented to provide transparency of decisions made.
- **4** The MoC should provide evidence that there has been active engagement and commitment from executives of organisations in the public and private sectors.
- 5 A communication strategy should be developed to promote awareness of the MoC development (and subsequently implementation) across the sector.

ESSENTIAL 🤸

What to ask and explore

Consultation outcomes should be defined and integrated into the final MoC.

2B

- **1** Throughout the development process, the MoC should document:
 - the consistency of stakeholders' perceptions/understanding of what needs to be done for implementation, particularly views on the urgency to change the current service model(s)
 - awareness amongst service providers of current problems with service delivery and willingness to work toward a solution
 - whether stakeholders are ready to work together, change resourcing patterns and/or relinquish control of certain services
 - the level of engagement from primary care service providers with the proposed MoC.
- 2 The ability to reach the target population and determine their acceptability/interest related to the MoC needs to be evidenced by mixed-methods research approaches (i.e. incorporating qualitative and quantitative approaches).
- **3** The MoC should outline current practice behaviours of clinicians and consumers and identify which components could be targeted to achieve improved outcomes.

Seeking endorsement

The MoC should be publicly endorsed by clinical and other organisational stakeholders as far as is practical and socially appropriate for local settings.

Principles:

- 1 Where feasible, the MoC should be endorsed and actively advocated by influential individuals and organisations, including consumer organisations. However, the process of attracting and confirming endorsement should not be a barrier to initiating implementation efforts.
- 2 The MoC should receive commitment from organisations to implement strategies (as evidenced through organisation-specific action plans, strategies, position statements), which are consistent with the recommendations in the MoC.
- **3** The MoC should be 'signed off' by an authorising agency, for example a government department.

2D Identifying and supporting local champions

The MoC development process should identify, engage and actively collaborate with local champions, where appropriate for the model and the environment in which it is to be implemented.

- 1 The MoC should identify local champions across relevant sectors in the community to encourage and support development and implementation of the MoC.
- **2** Champions should be both clinical and non-clinical (e.g. consumers, managers, administrators, funders).
- **3** The champions tasked with spearheading the MoC should be up-skilled in implementation science or change leadership in order to act as effective change agents.

3.

3A

Align to contemporary standards

ESSENTIAL 🤸

The MoC should align with standards of care for quality and safety and best practice for specific musculoskeletal health conditions. Best practice should be based on contemporary evidence and emerging reliable evidence for improved consumer and system outcomes.

Principles:

- 1 A MoC should outline and/or cite the quality and safety standards related to specific musculoskeletal conditions (where those standards are concordant with current evidence) and include strategies to mitigate quality and safety risks (e.g. time to surgery for hip fracture).
- 2 The MoC should be explicit about best practice across the care continuum, describing *what* the appropriate care is (based on evidence or best practice) and *how* it should be delivered effectively and efficiently.
- **3** In addition to addressing end stage disease and tertiary hospital activity, a musculoskeletal MoC should also consider service delivery in primary care and early disease identification and management as priorities.
- **4** The MoC should advocate for psychosocial assessment and intervention as part of service delivery.
- **5** The MoC should prioritise community care over tertiary hospital care, where appropriate.
- **6** The MoC should include strategies to optimise transition services for adolescents from paediatric to adult services.

3B Identify required behaviour changes

The MoC should clearly identify behaviour change priorities across stakeholders (as known at pre-implementation, recognising that a comprehensive set of priorities will not be realised until implementation has commenced).

- Behaviour change recommendations in the MoC should be informed by qualitative research to understand current local practice behaviours and barriers to practice change at the provider, administrator and consumer levels.
- 2 Behaviour change recommendations should be prioritised and supported by a theoretical model/framework of behaviour change (where relevant to 'real world' practice), such as the Behaviour Change Wheel¹⁹, or make reference to local case studies where sustainable behaviour changes have been observed.

Utilise different service delivery modes

The MoC should consider different delivery modes known to be effective to reach the target population, while maintaining fidelity to any critical aspects of the MoC.

3C

Principles:

- 1 The MoC should consider:
 - service providers who can cater to culturally and linguistically diverse groups
 - provision of information/resources in different languages and targeted to different age groups (e.g. adolescents)
 - information, communication and resource delivery channels that are accessible to people with disabilities and those who live in rural and remote areas (e.g. telehealth, social media).

3D Specify communication and referral pathways

The MoC should describe an ideal consumer journey through the system and suggest referral and communication pathways between service providers to facilitate a seamless journey for the consumer and their information, recognising that these pathways may need iteration during implementation.

- Ideal communication and referral strategies between providers and organisations should be described with explicit mention of transition processes between services/organisations.
- **2** The MoC should consider how care should be delivered where musculoskeletal health conditions are comorbid with other chronic health conditions.
- **3** Ideally, multidisciplinary services for consumers should be co-located or digitally connected to minimise travel burden for consumers/families.

Practical, user-friendly recommendations

ESSENTIAL 🤸

The MoC should be consumer-centred in all aspects and user-focused when describing recommendations for implementation.

4A

Principles:

- 1 The MoC should use language that is consumer-centred and use terminology or concepts that are meaningful to typical end users (e.g. service managers/funders).
- 2 The MoC should not be too large, or have too many recommendations. It needs to be presented in a manner that is directly usable and have an accompanying implementation plan and/or guidance for the development of a detailed Model of Service Delivery or business plan for a local setting.
- **3** Recommendations for implementation may include a range of issues to enable improved consumer outcomes, e.g. changes to health service delivery practices, changes to funding models, research priorities, health professional training, development of consumer resources, and so on. The relative priority of these recommendations may change over time.
- **4** The MoC should outline an improved consumer flow/journey through the system compared to the status quo.

4B

Partnership-based service delivery and funding

The MoC should promote cross-sectoral pathways and priorities in service delivery and service funding (e.g. public-private partnerships).

- **1** The MoC should give examples and encourage development of effective, collaborative models of service delivery and service funding in line with its recommendations.
- **2** The MoC should be amendable to funding support from different sectors of the health system.

INITIATING IMPLEMENTATION STREAM

This stream describes how to approach implementation after a MoC has been developed. It provides guidance on what to consider for optimising implementation success and how to develop an implementation plan. This stream is relevant to those tasked with implementation of a MoC, usually at a local or regional level.



KEY DOMAIN

5. Optimising implementation and evaluation success

5.

5A

OPTIMISING IMPLEMENTATION AND EVALUATION SUCCESS

Assess system readiness

The health system is dynamic and as such it is recognised that it will never be completely 'ready' for implementation since the system is constantly evolving. Therefore, it is important to identify critical components in the system that will impact on implementation efforts.

Principles:

The MoC should identify any local, jurisdictional or national systems, infrastructure, resources (e.g. funding, workforce) and processes that need to be considered to support implementation of the MoC and its evaluation at a local level; ideally evidenced by a system audit (refer to section 1B). Guidance should be provided on:

- what systems/resources are needed for successful implementation, in particular funding models and existing workforce volumes and distributions
- their state of readiness to support the MoC, as informed by thorough consultation with the local sector
- strategies to ensure sustainability of the necessary system modifications.

5B Linking to local resources

The implementation plan should contain, or link to, local resources to support implementation of specific components of the MoC.

- 1 The implementation plan should specify whether specific business plans or detailed service delivery models are required for particular components of the MoC. Alternatively, a framework for such business plans/detailed service models for end users could be included.
- 2 The implementation plan should include practice enablers to maximise uptake in target environments e.g. decision tools, care pathways/algorithms available to clinicians. Enablers should consider the recognised components of behaviour change¹⁹.
- **3** Where possible, local implementation support should be available to assist sites to implement a MoC successfully. Possible examples include:
 - web portal with online implementation training
 - assistance with trouble shooting
 - phone support to a coordinating agency, group training, email distribution list for receiving updates/new tools to support implementation.
Identifying likely workforce requirements

The implementation plan should consider workforce implications and indicative capacity requirements, recognising that flexibility and continuous re-calibration are required as implementation progresses.

5C

Workforce volumes, distributions and activities should remain the decision and responsibility of individual health services.

Principles:

- 1 Local workforce requirements to support implementation to achieve effective, integrated care should be identified within the implementation plan including:
 - minimum competencies for safely and effectively delivering the recommended care
 - minimum service and performance standards for the workforce and delivery sites for dealing effectively with arthritis, osteoporosis, pain and effectively supporting consumers to engage in disease-appropriate self-management behaviours
 - existing frameworks to support workforce roles recommended by the MoC e.g. nurse practitioners, extended scope of practice roles and assistant roles
 - behaviour change strategies for primary care providers, particularly general practitioners, to support consumers to engage in co-care (with a focus on weight loss, physical activity and exercise).
- 2 The implementation plan should consider whether local staff are empowered to implement system changes and whether they have the skills, knowledge and attitudes to do so. If not, strategies to address these potential barriers should be included, based on behaviour change principles (refer to 3B).
- **3** The implementation plan should include skills/knowledge training recommendations for key personnel involved in the implementation of the MoC.
- **4** A contemporary musculoskeletal MoC should also support up skilling of non-traditional primary care providers e.g. pharmacists, fitness instructors etc. in developed countries and relevant care providers in developing countries.
- **5** Outreach services should be inclusive of training for local clinicians in order to build local workforce capacity (e.g. the World Spine Care initiative⁷).

Building a comprehensive implementation plan

Once the MoC document has been completed, an implementation plan should be developed which includes guiding principles to inform the development of a locally relevant project, or business plans to facilitate implementation of specific components of the MoC.

5D

Principles:

- 1 The implementation plan should be based on, or supported by:
 - results from local pilot studies that identified and mitigated barriers and enablers (formative evaluation)
 - planned or opportunistic partnerships between service providers and responsiveness to current opportunities and contexts
 - a specific implementation theory/framework, particularly as it relates to supporting behaviour change of those people involved in implementing the MoC (such as the Behaviour Change Wheel¹⁹)
 - a formal change management process.
- **2** The implementation plan should:
 - include key performance indicators (consumer and system-relevant), as recommended by the MoC developers
 - include an outline of cost or resourcing requirements (e.g. units of each resource) for the various implementation phases, relevant to the system and the consumer
 - allow some flexibility in the MoC implementation, particularly for rural and remote areas
 - describe the available or proposed funding models to support implementation of the MoC across different settings
 - include an ongoing plan for engagement with local stakeholders
 - identify specific components of the MoC that can be implemented in specific health sectors in a prioritised manner.
- **3** The implementation plan should clearly outline evaluation processes (e.g. data collection systems).
- 4 The implementation plan must recognise that updating will be required as new evidence, technologies and workforce roles evolve. In these circumstances, the developers should update the MoC.

Formative evaluation of MoC components

Formative evaluation of some service components of the MoC should be undertaken to demonstrate proof of concept, inform (but not prescribe) a larger-scale rollout and attract engagement and resourcing at scale.

5E

Principles:

- 1 Formative evaluation is important to identify implementation barriers, cost-effectiveness estimates and attract buy-in across the sector by demonstrating that the MoC is effective and adaptable across environments. However, formative evaluation outcomes should only be used as a guide, since scaling up implementation from a pilot to a large scale will require adaptability.
- **2** Pilot sites need to be varied and represent the larger scale, not just the sites/people who are ready adopters.
- **3** Formative evaluation must include key performance indicators (**KPIs**) that are that are identical to the planned KPIs for larger-scale roll-out, including economic analyses.
- **4** Information and evidence from formative evaluations or large-scale implementation of other comparable MoCs should be considered, where relevant.
- 5 Recognise that where only components of the MoC are piloted (based on identified priority components), the results of individual components may vary when a total MoC is implemented.

5F

Establishing a User Reference Group

A multidisciplinary User Reference Group should be established to oversee implementation planning and execution.

Principles:

- **1** The User Reference Group (**URG**) should be multidisciplinary (inclusive of consumers and carers) and ideally be clinically-led.
- **2** The URG should have executive-level endorsement to support changes to implementation approaches.
- **3** The URG should review implementation and outcomes over time and recommend changes to the MoC developers.
- **4** The URG should monitor and respond to upstream and downstream consequences of implementing the MoC, such as impacts on service access and efficiency in other services/clinical areas.

SUCCESS STREAM

This stream considers how to approach evaluation, including both formative evaluation and impact evaluation that includes consumer and system-relevant outcomes. This stream is relevant to those tasked with monitoring the outcomes of a MoC, usually at a local or regional level.



KEY DOMAINS

- 6. Continuous improvement processes
- 7. Key performance indicators
- 8. Engagement and participation
- 9. Uptake and integration

Pragmatic evaluations over time

6A

A pragmatic evaluation has been undertaken at different time points, inclusive of outcomes (impact) and process (formative) evaluations.

Principles:		Performance indicators/methods/data:	
	plan has been developed which outcomes (impact) and process valuations.	 Outcomes should measure to what extent components were implemented, or likely to be implemented in a specified time period. Qualitative and quantitative measures linked to key performance indicators identified during MoC development (see 1A). 	
mixed-metho	eds to be informed by ds approaches, rather than e on evidence from randomised RCTs) only.	 Qualitative methods. Quantitative methods - surveys, quality audits, economic modelling, RCTs. 	
consumer-rel	tcomes need to be evant, provider-relevant elevant and map to specific of the MoC.		
i. short-tern change a	tcomes should consider: n outcomes that reflect behaviour nd system efficiency improvements	• Short term outcomes: qualitative and quantitative data from clinicians and consumers, service activity outcomes.	
effectiven	rm outcomes should reflect the less of the behaviour changes ber of people who sustain es).	 Longer term outcomes: population-level health and system activity outcomes from jurisdictional health surveillance systems. 	
outcomes (po been achieve	n plan has the flexibility to capture ositive or negative) that have d which are not directly related nentation of the MoC, including chievements.	• Qualitative methods.	
settings is a p	lity of the MoC to different referred indicator of success entation fidelity.	• Qualitative methods to understand implementation fidelity.	

ESSENTIAL 🚽

Quality assurance and troubleshooting mechanisms

6B



The MoC has ongoing quality assurance and troubleshooting processes in place.

Principles:	Performance indicators/methods/data:	
1 Systems or strategies have been developed to address implementation 'hurdles'.	 Ongoing implementation risk assessment/audit. Qualitative methods to identify barriers/enablers to implementation. 	
2 Identification and documentation of "lessons" that can be applied to achieve more successful implementation of other MoCs or implementation in another setting.	• Qualitative assessments with stakeholders to identify barriers/enablers to implementation.	
3 There is flexibility in processes/programs to respond to emerging evidence for new technologies, programs or interventions; or experiences of users.	 Regular review/updates of peer-reviewed literature to identify new evidence. Qualitative assessments with stakeholders (interviews, focus groups) to identify changing practice behaviours. Audits are undertaken to ensure changes to the MoC have been made appropriately, e.g. via a User Reference Group (see 5F). 	
4 Regular audits of consumers' pathways of care are undertaken to ensure alignment with the MoC recommendations.	 Qualitative assessments with stakeholders (interviews, focus groups) to measure satisfaction with services. 	

Data collection processes for key performance indicators



Data collection processes have been established to measure pre-defined key performance indicators (**KPIs**), as well as KPIs that are later identified to be important.

Principles:

6C

 Consumer-relevant (e.g. quality of life) and system-relevant data collection processes have been implemented that measure and monitor the MoC impact through routine surveillance initiatives maintained by agencies.

Performance indicators/methods/data:

- Health surveillance registries.
- System activity databases.
- **2** As implementation progresses, further important key performance indicators (**KPIs**) are identified and are added or replace original KPIs.
- Qualitative assessments with stakeholders to identify barriers/enablers to implementation.

6D Promoting research priorities

The MoC identifies and promotes research priorities.

Consumer relevant outcomes

Over time, there is evidence of improved consumer experiences, access, health outcomes and quality of life.

Principles:

Λ'

- **1** Improved consumer **experiences** relating to services and pathways measured through:
 - Satisfaction with both access to, and quality of, healthcare services.
 - Improved health literacy.
 - Consistency of service quality and access across sites.
 - Improved self-management support from providers and shared decision-making between providers and consumers.
- 2 Improved **access** for consumers measured through:
 - Access to services and information close to home, where clinically appropriate.
 - Effective services, especially in ambulatory care and community settings.
 - Reduced inequalities to accessing care.
- **3** Long term consumer **health outcomes** should include:
 - Positive changes in function/participation and the experience of disability.
 - Improved quality of life.
 - Population health outcomes (measured as a long term outcome) e.g. joint replacement rates, obesity, re-fracture rates.

40 A FRAMEWORK TO EVALUATE MUSCULOSKELETAL MODELS OF CARE

Performance indicators/methods/data:

- Qualitative methods.
- Quantitative surveys with consumers/families.

- Qualitative methods and surveys with consumers/families.
- Service indicators such as:
 - waiting time to access services
 - episodes of multidisciplinary service uptake between groups with varying access inequality prior to implementation of the MoC.
- Standard PROMIS tools (where applicable).
- System indicators, such as: joint replacement rates, fracture/re-fracture rates and obesity prevalence.
- Data sources include routine jurisdictional health surveillance initiatives, population health registries.



Service delivery partnerships and pathways

There is evidence of improved consumer pathways and establishment of partnerships and workforce efficiencies (longer term outcome).

Principles:

7B

- Consumer pathways, as defined in the MoC, are implemented as evidenced through:
 - Patient flow e.g. timely and appropriate access to surgery.
 - Early identification/diagnosis.
 - Transition services for adolescents.
 - Timeliness of services (e.g. waitlist no longer than 6-12 weeks for early inflammatory arthritis review).
 - Patient safety outcomes.
 - Reduction in unnecessary referrals, interventions and diagnostic tests (e.g. imaging for non-specific back pain).
- 2 Improved service partnerships between hospitals and community services/primary care.

• Extended scope of practice or advanced

• Extent to which interprofessional practice

practice roles implemented.Professional development/training

initiatives commenced.

Performance indicators/methods/data:

- System activity metrics (e.g. referrals, access points, episodes of care for a given diagnosis).
- Qualitative methods with consumers/families and surveys.
- System activity data from data sources that track episodes of care for specific diagnoses such as the Electronic Persistent Pain Outcomes collaboration initiative.

- Stakeholder feedback via surveys and qualitative methods.
- Establishment of service or purchasing agreements that reflect recommendations in the MoC.
- Establishment of governance frameworks to support new workforce roles.
 - Appointments for specific workforce roles.
 - Development and evaluation of professional development resources/initiatives.
- Episodes of care data by discipline.

4 Changing service utilisation:

is being adopted.

3 Workforce efficiencies:

- Shift in services from tertiary to primary care.
- Appropriate evidence-based interventions being accessed.
- Healthcare utilisation information collected via consumer surveys.

Cost-effectiveness

Where reliable and sufficient data are available, the cost-effectiveness of the MoC is determined (estimated through modelling or measured in the long term) or cost utility evaluation undertaken.

Principles:

7C

Cost-effectiveness measured through:

- Unit cost per patient treated.
- Jurisdictional costs over time for specific conditions.
- Cost shifting from one sector to another.
- Disability adjusted life years (DALYs) saved or quality adjusted life years (QALYs) gained.
- Resources per patient outcome, inclusive of social care.
- Cost to the consumer.

- Performance indicators/methods/data:
- Jurisdictional or site-specific data on costs of service delivery, inclusive of public and private systems.
- Direct service costs to consumers.
- Healthcare utilisation data from consumer surveys.

Stakeholder behaviour changes



Once fully implemented, there is behaviour change amongst stakeholders, led initially by opinion leaders, aligned to the recommendations of the MoC.

Principles:

7D

- Practice behaviours among providers (led initially by local opinion leaders) and consumers related to the recommendations of the MoC are observed, including:
 - Enhanced care coordination by healthcare professionals across sectors.
 - Improved self-management/self-monitoring by consumers.
 - Increased interaction between administrators/executives and clinicians relating to service delivery.
 - Increased interprofessional practice across sites.
 - Changes to referral rates or other behavioural change metrics by discipline.
- 2 There is increased adoption of the MoC, as evidenced by increased behaviour change across the sector (clinicians and sites) – i.e. a 'ripple effect'.

Performance indicators/methods/data:

Examples of specific indicators:

- Number of health assessments.
- Review of prescriptions.
- Supporting physical activity as a primary intervention for chronic conditions such as osteoarthritis care.
- Data methods include audits and surveys.
- Audits of the consumer pathway.
- Data characterising episodes of care by discipline.

8A

Awareness and knowledge of the MoC

There is an awareness of the MoC amongst stakeholders and organisations (inclusive of consumers) in the long term.

ESSENTIAL

Principles:	Performance indicators/methods/data:
 There is an awareness of the MoC or its recommended services across a broad range of stakeholders, including consumers/carers. 	 Surveys/audits among stakeholders: % heard of MoC/its services % aware of MoC/its services % read MoC. Qualitative methods supported by quantitative outcomes, where feasible.
 2 Assessment of knowledge and attitudes of service providers, administrators and consumers relating to: Understanding of how their role fits in as part of the MoC. Working effectively to support broad 	• Qualitative methods (interviews, focus groups), supported by quantitative outcomes, where feasible.
direction of MoC.Support/commitment to the MoC.Extent to which the MoC is part of	
day-to-day activities.3 Community awareness of the impact of musculoskeletal health conditions.	 Policy statements about population health related to musculoskeletal conditions.
4 Uptake of resources for consumers.	 Utilisation of non-government organisations' resources.

Reach to target population

A cohort of the target population for which the MoC was developed has interacted with the MoC.

Principles:	Performance indicators/methods/data:	
1 Proportion of target population that the MoC has reached through programs, activities or other specified modes in the MoC are measured.	 Utilisation of services for different musculoskeletal conditions. Comparison of patient demographics compared to broader target population demographics. 	
2 The population that has been 'reached' by the	 Consider demographics relating to age, 	

- 2 The population that has been 'reached' by the MoC is comparable with known demographics of the target population to ensure representativeness.
- Consider demographics relating to age, gender, location (urban, regional, rural), racial diversity, diagnoses.

8C Satisfaction with processes and programs

There is evidence of satisfaction among service providers and consumers with the MoC and its implementation compared with service delivery pre-implementation, as informed through ongoing consultation with stakeholders.

Principles:

8B

- There is evidence of satisfaction among service providers and consumers with processes and programs associated with the MoC and/or analysis to understand reasons for dissatisfaction.
- **2** There is increased confidence among service providers with the services provided to consumers.

Performance indicators/methods/data:

- Surveys with stakeholders and qualitative methods.
- Surveys with stakeholders and qualitative methods.

9A

Adaptability across settings and responsiveness ESSENTIAL

While retaining its core features, the MoC has adaptability to be implemented in different contexts/environments/cultures and evolve over time as new evidence, technologies or workforce roles emerge, or as contextual demands require.

Principles:		Performance indicators/methods/data:	
1	Stakeholders or organisations have adapted the MoC or its components to their settings, while maintaining fidelity to its core features.	 Proportion of locations/organisations that have implemented the MoC as a whole and/or in components. 	
2	The extent of adaptability will depend on the system-level focus of the MoC - national, regional, or local; developing or developed countries.	 Proportion of core elements/components/ programs of the MoC that have been consistently implemented across different environments. 	
3	The MoC should be viewed as a 'living resource' that can be updated over time as new evidence, technologies and workforce roles emerge or experiences with implementation efforts (e.g. implementation failures and successes) emerge.	• Identification of where implementation of the MoC has deviated and whether sites are still achieving outcomes consistent with the objectives of MoC.	

Innovative changes to service resourcing

9B

Innovative service provision and workforce models associated with the MoC are implemented.

Principles:	Performance indicators/methods/data:	
 Changes in resource allocations (e.g. workforce roles) are observed to support the MoC implementation. 	Qualitative methods.Service audits.	
2 Service activities relevant to the MoC are being resourced or adequately provided when resources are redistributed.	 Policy/contractual changes related to funding arrangements. 	
3 Resourcing for services/activities discordant with the MoC have been reduced and reallocated to services/activities aligned to the MoC.	• Frameworks available to support workforce roles aligned to the MoC.	
4 The MoC becomes part of service agreements or activity-based funding arrangements.	 Improved purchasing or service agreements. 	

9C The MoC becomes routine business

The MoC becomes part of routine business for the sponsor organisation and other organisations in the longer term.

Principles:		Performance indicators/methods/data:	
1	Organisational position statements or strategies have changed to align with the MoC.	• Review of organisational strategies/business plans.	
2	Key elements of the MoC have been incorporated into the core business of target organisations (e.g. strategy, business plans).	• Review of organisational strategies/business plans.	
3	The MoC is considered as standard of care for organisations.	• Review of organisational strategies/business plans.	

9D The MoC is utilised as a resource

The MoC is being used as a resource across the sector to communicate best practice care.

Principles:	Performance indicators/methods/data:	
1 Practice behaviours aligned to the MoC have been integrated into the curriculum of universities and other training organisations.	• Surveys and audits.	
2 The MoC has been referenced in government and research publications or public reports.	• Citations of the MoC in reports or publications.	
3 The MoC is used/referenced to develop policies, strategies or clinical guidelines in organisations.	• Recognition of the MoC in health policy.	
4 The MoC recommendations have been incorporated into discipline-specific guidelines, in particular primary care guidelines.	• Recognition of the MoC in clinical guidelines.	
5 The MoC has links to best practice consumer resources.		

9E) The new MoC replaces the old MoC

There is evidence of the MoC replacing previous MoC/practices, where appropriate.

Principles:

1 Practices that are not consistent with the new MoC have been discontinued.

Performance indicators/methods/data:

- Data methods include surveys, audits of health services.
- Qualitative and survey-based research with consumers who are interacting with health services.



PUTTING THE FRAMEWORK INTO PRACTICE



This section provides examples of how the Framework could be applied in practice. While the examples are based on musculoskeletal conditions, they are equally relevant to other non-communicable diseases.

Three scenarios are described, which demonstrate application of the three streams of the Framework. Following each scenario is a schematic of the process described.



EXAMPLE SCENARIOS

- 1. Using the Framework to judge the readiness of a Model of Care for osteoarthritis
- 2. Using the Framework to assess preparedness for implementation or prepare for implementation
- 3. Using the Framework to judge the success of a Model of Care for osteoporosis

SCENARIO 1: USING THE FRAMEWORK TO JUDGE THE READINESS OF A MODEL OF CARE FOR OSTEOARTHRITIS

The Department of Health in a particular jurisdiction has identified the need to address the rising burden of disease associated with hip and knee osteoarthritis (**OA**) in the community. An expert group of clinicians and other stakeholders have developed a MoC for osteoarthritis and are keen to implement it.

Given the limited resources for new health services and programs and diverse views and practices on the management of OA, the development group is concerned about how the MoC will be accepted and supported by the local healthcare sector. The group refers to the Readiness stream of the Framework to benchmark the content of their MoC and the processes undertaken to develop it.

From a process perspective, the Readiness stream can be considered in three phases:

- **1.** Target and address a local health issue (Figure 7) inclusive of:
 - 1.1 defining and describing the problem
 - **1.2** mapping the evidence
 - **1.3** developing preliminary recommendations.
- **2.** Establish a consultation network (Figure 8).
- **3.** Deliver a Model of Care document based around a consumer's journey (Figure 9).

Phase 1: Target and address a local health issue

1.1: Defining and describing the problem

- ✓ Clearly define the local health issue to be addressed - in this case hip and knee OA.
- Describe the problem based on local and national data and issues of importance to local consumers and service providers.
- ✓ Inform this 'case for change' based on thorough consultation with a range of stakeholders (e.g. consumers/carers, clinicians, consumer advocacy group, policy makers, researchers, service providers).

1.2: Mapping the evidence for addressing the health problem, in this case OA. Describe *what* care should be provided and *how* it should be delivered.

Relevant evidence is likely to include a mix of the following, according to what is available and feasible locally:

- Established standards for service safety and quality for OA (e.g. those produced by eumusc.net and reported recently).²⁰
- Beliefs and attitudes of local service providers regarding OA co-care (e.g. important components of a care pathway).
- Searching for what has worked elsewhere (e.g. other countries) and what might be some future developments (horizon scan) in therapies, workforce roles, programs or interventions for OA e.g. physiotherapy triage roles.
- Best practice based on high-level research evidence (for example, meta-analyses, systematic reviews, randomised controlled trials) for OA.
- Local audits and experience to identify what works locally for OA.
- Describe how the consumer journey might be improved, particularly as it relates to pathways between services and the important components of care (e.g. process of referral from primary care to other service providers).

1.3: Developing preliminary recommendations

For example:

- All patients diagnosed with OA should receive education about exercise and weight loss.
- Multidisciplinary teams should manage advanced OA.



Figure 7: Schematic of Readiness Stream Phase 1 involving targeting a local health issue, mapping evidence and developing preliminary recommendations.

Phase 2: Establish a consultation network

- Identify and engage a broad range of stakeholders across the health sector, inclusive of consumers and carers, who can comment on:
 - the consumer pathways described for OA
 - recommendations for service and resource improvements (see 1.3).
- Where relevant, stakeholders of different ages, socioeconomic status and geographies should be consulted.
- The recommendations should specify *what* care should be provided and *how* it should be delivered.
- Identify and engage with local champions who can help drive local consultation efforts.
- Consult with these stakeholders on the draft MoC for OA and iterate according to feedback.
 For example, stakeholders may seek further clarification on workforce roles.



Figure 8: Schematic of Readiness Stream Phase 2 involving establishing a consultation network and iteratively refining recommendations for the MoC.

PART 4: PUTTING THE FRAMEWORK INTO PRACTICE 53

Phase 3: Deliver a Model of Care document based around a consumer's journey

- The final Model of Care document for OA is based on the cumulative output of Phases 1 and 2 and framed around a consumer journey.
 - For example, the Model of Care describes components of care from diagnosis to end-stage disease management, inclusive of resources and interventions to assist the person with OA.
- Ideally, the document should contain specific components:
 - Aims and scope.
 - Consumer KPIs such as access to care, wait times.

- System-relevant KPIs such as cost per episode of care.
- Data-driven case for change such as metrics around prevalence, impact and costs of care.
- A locally-informed commentary around current systems and behaviours.
- A commentary around how the proposed MoC aligns with jurisdictional health policy, chronic disease management frameworks etc.
- Clear recommendations, prioritised for implementation, describing *what* care and *how* it should be delivered.
- The Model of Care should ideally be publicly endorsed by an authorising agency (e.g. government department) and a consumer version should be made available.



Figure 9: Schematic of Readiness Stream Phase 3 outlining recommended components of the final MoC document.



Figure 10: Schematic of Readiness Stream Phase 3 outlining recommended components of the final MoC document.

SCENARIO 2: USING THE FRAMEWORK TO ASSESS PREPAREDNESS FOR IMPLEMENTATION OR PREPARE FOR IMPLEMENTATION

A MoC has been developed to improve care for people living with chronic back pain. Having considered the Readiness stream of the Framework and deemed the development process for the MoC to have been optimal, the developers now wish to prepare for its implementation within their local health system.

Initiating implementation of the MoC is primarily based on developing an appropriate implementation plan for local users.

This is informed by a formative evaluation/pilot studies (**Phase 1**) to assess the local feasibility of the MoC and key performance indicators identified for evaluation purposes (see Figure 11).

The implementation plan should provide guidance for the development of detailed business cases or project management plans to enable local users to implement components of the MoC (Phase 2).

- **Phase 2A:** Prepare an implementation plan (Figure 12).
- Phase 2B: Prepare local business cases or project management plans (Figure 13).

Phase 1: Formative evaluation

After developing the MoC, a formative evaluation is undertaken to consider:

- The development process of the MoC.
- Any likely barriers and enablers to implementation.
- Cost modelling related to the MoC.
- The feasibility of key performance indicators.

This information informs the development of an Implementation Plan for the MoC.



Figure 11: Elements of formative evaluation to assess local feasibility of the MoC

Phase 2: Enable local users to implement components of the MoC

Phase 2A: Prepare an implementation plan

The Implementation Plan should consider:

- Judging the likely readiness of the local health system to implement the MoC:
 - particularly as it relates to the will of the sector, infrastructure and resourcing and workforce capacity requirements (e.g. appropriate training in pain management).

- Determine requirements for data collection systems.
- Partnership opportunities (e.g. conjoint employment between a hospital and community health service).
- Costs required for implementation (e.g. salary triage officers).
- A plan for ongoing consultation with stakeholders.
- Further iterations of the Plan based on consultation with stakeholders and findings from formative evaluations.



Figure 12: Schematic of Phase 2A of the Initiating Implementation Stream illustrating components in the development of an implementation plan.

Phase 2B: Prepare local business cases and project management plans

The Implementation Plan is used to inform detailed operational plans for local settings in the form of business cases and project management plans.

• For example, the establishment of a local triage process for people presenting to emergency departments with back pain.

The implementation of the MoC is likely to be most successful when local business and project management plans consider and establish key enablers including:

- Data collection systems, e.g. systems to track pain and disability outcome measures for consumers with back pain.
- Clinical tools, e.g. imaging/opioid guidelines for general practitioners.

- Upskilling strategies for the workforce (e.g. professional development opportunities related to self-management support) and consumers (e.g. consumer website about effective pain management options).
- A communication strategy about the MoC to inform local stakeholders.
- Support for implementation by a central agency or User Reference Group.
- Guiding principles described in the Implementation Plan.
- Clearly outlined indicators to measure effectiveness (e.g. function for consumers and reduced system costs for misuse of imaging/opioids).



PHASE 2B PREPARE LOCAL BUSINESS CASES AND PROJECT MANAGEMENT PLANS

Figure 13: Schematic of Phase 2B of the Initiating Implementation Stream focusing on the development of local business cases and project management plans, informed by the Implementation Plan.



Figure 14: Composite schematic of Initiating Implementation Stream

SCENARIO 3: USING THE FRAMEWORK TO JUDGE THE SUCCESS OF A MODEL OF CARE FOR OSTEOPOROSIS

A MoC for osteoporosis has been implemented across tertiary hospitals in a health area for a number of years.

The MoC is based around identification and management of minimal-trauma fractures using a Fracture Liaison Service. The service was identified as a strategy by local stakeholders and informed by international best practice models^{21,22}. Having supported the MoC for a number of years, health administrators now wish to understand whether the MoC has been successful.

Having established local data collection systems as part of the implementation preparation phase (refer to scenario 2), impact evaluations are now undertaken. Three components should be considered in the judgement of success:

- 1. Undertaking impact evaluation, inclusive of:
 - **1.1** Measurement of system-relevant indicators of success.
 - **1.2** Measurement of consumer indicators of success.
- 2. Ongoing implementation quality assurance.
- 3. Sector-based indicators of success.

1: Establishment of local data collection systems for impact evaluation

- As part of the Initiating Implementation Stream, local data collection systems and processes should be established (refer to scenario 2).
- These systems should be able to measure system and consumer-relevant indicators of success through pragmatic, mixed-methods approaches in impact evaluations.

1.1: Measurement of system-relevant indicators of success

- Partnerships, e.g. between hospitals and osteoporosis support organisations.
- Costs, e.g. system costs for re-fracture admissions²³ and orthopaedic surgeries.
- Service use, e.g. re-admissions for fractures.
- Workforce roles, e.g. establishment of Fracture Liaison Services across sites.
- Behaviour changes, e.g. screening and management for osteoporosis after minimal trauma fracture.
- MoC in routine business, e.g. the extent to which Fracture Liaison Services are implemented across a system in a recurrent manner.

1.2: Measurement of consumer outcomes

- Consider consumers' experience with health services and components of care on the service pathway (e.g. wait time to surgery).
- Ease of access to services (e.g. access to bone densitometry).
- Consumer-relevant health outcomes (e.g. measured via survey).



Figure 15: Schematic of component 1 of the Success Stream identifying relevant system and consumer-relevant indicators of success.

Component 2: Ensure ongoing implementation quality assurance

- Establish a multidisciplinary and clinically-led User Reference Group (**URG**) (refer to Initiating Implementation Stream, Figure 14) to support implementation efforts.
- The URG should update the Implementation Plan and MoC based on:
 - new evidence
 - feedback from stakeholders
 - audits of the consumer pathway
 (e.g. from fracture presentation to screening and management of osteoporosis).

Component 3: Determine whether sector-based indicators of success have been observed

Relevant indicators may include:

- Awareness of the MoC (e.g. fracture liaison services) by local providers.
- The MoC is used as a resource (e.g. for service planning at other sites).
- Identifying research priorities (e.g. long term cost-effectiveness of Fracture Liaison Services).
- Satisfaction among stakeholders (e.g. referrals from emergency departments to fracture clinics).
- Adaptability of the MoC to different settings, such as rural settings.²⁴
- Evidence that the target population has benefited (e.g. the re-fracture rates rate per head decline in the long term).





Figure 16: Schematic of components 2 and 3 of the Success Stream outlining a User Reference Group (URG) to support implementation quality assurance and indicators of sector-based success of the MoC.

COMPONENT 1 ESTABLISHMENT OF LOCAL DATA COLLECTION SYSTEMS







Figure 17: Composite schematic of Success Stream



SUPPORTING INFORMATION

Champion: influential individuals who are recognised opinion leaders or 'change agents' in their community or profession, or individuals who are invested in health reform initiatives and can advocate effectively for change.

Co-care: refers to a co-operative approach to care delivery and evaluation undertaken between a health professional(s) and the consumer of the care, together with their family or guardians.

Consumer: any actual or potential recipient of healthcare (e.g. patient in hospital, client in a health centre, person seeking health information via the Internet). In this context, the term 'consumer' does not relate solely to a monetary transaction for a health service.

Cost-effectiveness: analysis that compares the relative costs and outcomes of different Models of Care. In this way, cost is balanced against health outcomes.

Consumer outcome/consumer-centred: an outcome relevant to a consumer of a health service or resource, such as a patient. Consumer-centred outcomes typically include function, quality of life and satisfaction with health services.

Cost shifting: moving or reallocating a cost from one part of the health system to another. Therefore, cost shifting does not infer a cost saving.

Disability Adjusted Life Year (DALY): A measure of overall disease burden, expressed as the number of years lost due to ill-health, disability or early death.

Delphi: a research method involving a structured communication approach with a panel of subject matter experts to explore issues around a topic and sometimes reach a consensus.

End user: an individual or organisation who uses a resource or product after it has been fully developed.

Formative evaluation: a method for judging the worth of a program, while the program activities are forming (in progress). Formative evaluation assesses program design, early implementation and associated outcomes and is generally undertaken before a program is implemented across a system. A formative evaluation is sometimes conducted on a pilot of a Model of Care in a selected site(s), or on the first phase of implementation. The data and findings from a formative evaluation can be used:

- As the basis to determine the impact that a Model of Care might have if it was systematically implemented.
- To further refine the Model of Care by establishing early outcomes arising from programs and to identify areas requiring improvement.
- To enhance the probability of achieving program outcomes in the short, medium and longer term²⁵.

Framework: a structure that underlies and integrates a set of concepts or principles in a meaningful way to allow use by others.

Horizon scan: a non-systematic review, informed by experts, of likely future initiatives that will influence the Model of Care. Examples include new therapies for treating diseases, new workforce roles, new programs or policies.

Impact evaluation: an assessment of how the intervention being evaluated influences outcomes, whether these effects are intended or unintended. An impact evaluation forms part of a summative evaluation which assesses the quality, outcomes and impact of a Model of Care²⁵.

Implementation science: the study of methods to promote the uptake of research findings into routine healthcare policy and practice.

Key Performance Indicator (KPI): metric used to define and measure progress towards achieving an objective, outcome or goal. KPIs are objective and quantifiable and often time-based.

Model of Care (MoC): is a person-centred and principle-based guide, usually presented as a document, that describes evidence-informed, best practice care for particular health conditions (in this case, musculoskeletal conditions). It outlines *what* care should be provided and *how* it should be delivered. A Model of Care is not an operational plan for a health service or a clinical practice guideline. MoCs are usually implemented as health services at a local level.

Non-communicable disease (NCD): condition or disease that is non-infectious or non-transmissible. NCDs usually refer to chronic diseases that last for long periods of time and progress slowly.

Practice enablers: tools, strategies, resources or systems that equip providers to deliver best practice care.

Patient Reported Outcomes Measurement Information System (PROMIS): It is a system of highly reliable, precise measures of patient-reported health status for physical, mental, and social well-being (see http://www.nihpromis.org/about/overview).

Quality Adjusted Life Year (QALY): A generic measure of disease burden, including both the quality and the quantity of life lived. It is used in assessing the value for money of a health intervention and forms part of a cost utility analysis.

Qualitative methods: a research approach that aims to explore in detail an audience's range of behaviours and the perceptions that drive them, with reference to specific topics or issues. A range of methods can be undertaken, such as interviews and focus groups.

Readiness: the extent to which a Model of Care is ready for implementation, based on specific development processes.

Randomised Controlled Trial (RCT): a gold standard research design to assess the effectiveness of an intervention where people being studied are randomly allocated to either a group that receives an intervention or to a comparison group.

Subject matter expert: a person who is an authority in a particular area or topic.

Success: the extent to which the desired consumer and system-relevant outcomes have been achieved after a Model of Care has been implemented.

System outcome: an outcome relevant to the health system function or governance. The outcomes generally relate to service use, service funding and workforce.

Years Lived with Disability (YLD): a measure of disability or morbidity that characterises the number of years of productive life lost due to disability. For further information, see: http://www.who.int /healthinfo/global_burden_disease/metrics_daly/en/.

Financial support

Funding to undertake the project was awarded by the Physiotherapy Research Foundation (Australia); Department of Health, Government of Western Australia; New South Wales Agency for Clinical Innovation (Australia); Liverpool Hospital, South Western Sydney Local Health District (Australia) and the School of Physiotherapy and Exercise Science, Curtin University (Australia).

Non-financial support

In kind support was provided by the Department of Health and Human Services, Victorian Government (Australia); HealthSense (Australia) Pty Ltd; and the Global Alliance for Musculoskeletal Heath, Cornwall, United Kingdom.

The project team appreciatively acknowledge Dr Sue Brennan, Australasian Cochrane Centre for providing expert advice for phase 3 of the project.

The following individuals are gratefully acknowledged for participating in all three rounds of the Delphi phase and have consented to their names appearing in this report. The project team sincerely thank them and the 13 unnamed experts for sharing their time and expertise.

Associate Professor J. Haxby Abbott,

University of Otago, New Zealand

Dr Kelli Allen, University of North Carolina and Durham VA Medical Center, United States of America

Dr Haseebullah Ammad, PSRD College of Rehabilitation Sciences, Pakistan

Mr Phil Baker, United Kingdom

Mrs Sarah Bazin, World Confederation for Physical Therapy (European Region), United Kingdom

Professor Kim Bennell, University of Melbourne, Australia

Dr Florian Berghea, Carol Davila University, Romania

Dr Thomas Best, Ohio State University, United States of America

Ms Dianne Bianchini, Department of Health, Government of Western Australia

Ms Lara Bloom, United Kingdom

Ms Lesley Brydon, PainAustralia

Ms Ainslie Cahill, Arthritis Australia

Ms Antonella Cardone, The Work Foundation, United Kingdom

Professor Loreto Carmona, Instituto de Salud Musculoesquelética, Spain

Dr Stephanie Davies, Painless, Australia

Dr Aileen M. Davis, University of Toronto, Canada

Dr Jacqueline Detert, Charité - Universitätsmedizin Berlin, Germany

Dr Luis R. Espinoza, LSU Health Sciences Center, United States of America

Mr Jonathan Field, Royal College of Chiropractors, United Kingdom

Ms Karen Filocamo, Health Consumers NSW, Australia

Dr Jacquie Garton-Smith, Department of Health, Government of Western Australia

Professor Elizabeth Geelhoed, University of Western Australia, Australia

Dr Mark Gurden, Royal College of Chiropractors, United Kingdom

Professor Scott Haldeman, University of California, United States of America

Professor Jan Hartvigsen, University of Southern Denmark, Denmark

Mrs Renee Hayden, Primary Health Networks, Australia

Professor Rana Hinman, University of Melbourne, Australia

Mr Ben Horgan, Western Australian Primary Health Alliance, Australia

Mr Nigel Hunt, Royal College of Chiropractors, United Kingdom

Professor David Hunter, University of Sydney, Australia

Ms Anna Huska, Department of Health, Government of Western Australia

Ms Marani Hutton, Department of Health, Government of Western Australia

Ms Katti Kõrve, East Tallinn Central Hospital, Estonia

Princess Rosemary Kirungyi, Consumer, Uganda

Professor Deborah Kopansky-Giles, World Federation of Chiropractic, Canada

Dr Mahendranath Kurugodu, Samarpan Health Centre, India Associate Professor Peter Larmer, Auckland University of Technology, New Zealand

Dr Ing Soo Lau, Selayang Hospital, Malaysia

Professor Robert Lindsay, Columbia University, United States of America

Dr Elizabeth Lingard, North East Quality Observatory Service, United Kingdom

Professor Stefan Lohmander, Lund University, Sweden

Dr Lise Lothe, Norway

Dr Brian Lucas, Queen Elizabeth Hospital, United Kingdom

Professor Ghassan Maalouf, International Fragility Fracture Network, Lebanon

Dr Roderic MacDonald, British Institute of Musculoskeletal Medicine, United Kingdom

Ms Kerry Mace, Consumer, Australia

Professor Lyn March, University of Sydney, Australia

Ms Linda Martin, Arthritis and Osteoporosis Victoria, Australia

Ms Rhona McGlasson, Bone and Joint Canada and Canadian Orthopaedic Foundation, Canada

Mr Paul Mitchell, University of Notre Dame, Australia/New Zealand

Ms Cathie Morton, Arthritis New Zealand, New Zealand

Ms Amanda Mulcahy, Australian Commission on Safety and Quality in Health Care, Australia

Professor Norimasa Nakamura, Osaka Health Science University, Japan

Dr Bobby Ng, World Federation of Occupational Therapists, China

Mr Clement Nhunzvi, University of Zimbabwe, Zimbabwe

Dr Alan Nye, Pennine MSK Partnership, United Kingdom

Dr Stathis Papadopoulos, East Mediterranean Middle East Chiropractic Federation, Cyprus

Ms Sigrid Patterson, New South Wales Agency for Clinical Innovation, Australia

Ms Marilyn Pattison, World Federation of Occupational Therapists, Australia

Professor Kevin Pile, Western Sydney University, Australia

Dr Carlos Pineda, Pan-American League Against Rheumatism and Instituto Nacional de Rehabilitacion, Mexico

Dr Nicola Plastow, Stellenbosch University, South Africa

Dr Daiana Popa, Romanian Society of Rehabilitation Medicine, Romania

Professor Johannes J Rasker, University Twente Enschede, The Netherlands

Dr Anne-Christine Rat, Université Henri Poincaré, France

Dr Tim Raven, European Council on Chiropractic Education, Norwegian Chiropractors Association, Norway

Professor Jean Yves Reginster, University of Liege, Belgium

Ms Chiara Retis, Handicap International, Nepal

Dr Neil Sheth, University of Pennsylvania, United States of America

Mr Mark Slattery, Department of Health, Government of Western Australia

Professor Tanja Stamm, Medical University of Vienna, Austria

Ms Michaela Stoffer, Medical University of Vienna, Austria

Dr John Tanner, British Institute of Musculoskeletal Medicine, United Kingdom

Adjunct Professor Simon Towler, Fiona Stanley Hospital, Australia

Associate Professor Gabriela Udrea, "Dr I Cantacuzino" Clinical Hospital, Romania

Mr Raj Verma, New South Wales Agency for Clinical Innovation, Australia

Ms Kim Watkins, University of Western Australia, Australia

Ms Belinda Whitworth, South Metropolitan Health Service, Western Australia

Professor Simon Willcock, Macquarie University, Australia

Dr James Witter, National Institute of Arthritis and Musculoskeletal and Skin Diseases, National Institutes of Health, United States of America

Dr Linda Woodhouse, Canadian Physiotherapy Association, Canada

- Vos T, Barber RM, Bell B, et al. Global, regional, and national incidence, prevalence, and years lived with disability for 301 acute and chronic diseases and injuries in 188 countries, 1990-2013: a systematic analysis for the Global Burden of Disease Study 2013. Lancet 2015;386:743-800.
- Briggs AM, Cross MJ, Hoy DG, et al. Musculoskeletal health conditions represent a global threat to healthy aging: A report for the 2015 World Health Organisation World Report on Ageing and Health. Gerontologist 2016;56:S243-S255.
- Davidson P, Halcomb E, Hickman L, Phillips J, Graham B. Beyond the rhetoric: what do we mean by a 'model of care'? Aust J Adv Nursing 2006;23:47-55.
- Briggs AM, Towler SC, Speerin R, March LM. Models of care for musculoskeletal health in Australia: now more than ever to drive evidence into health policy and practice. Aust Health Rev 2014;38:401-405.
- Speerin R, Slater H, Li L, et al. Moving from evidence to practice: Models of care for the prevention and management of musculoskeletal conditions. Best Pract Res Clin Rheumatol 2014;28:479-515.
- MacKay C, Veinot P, Badley EM. Characteristics of evolving models of care for arthritis: a key informant study. BMC Health Serv Res 2008;8:147.
- Haldeman S, Nordin M, Outerbridge G, et al. Creating a sustainable model of spine care in underserved communities: the World Spine Care (WSC) charity. Spine J 2015;15:2303-2311.
- Hoy D, Geere JA, Davatchi F, Meggitt B, Barrero LH. A time for action: Opportunities for preventing the growing burden and disability from musculoskeletal conditions in low- and middle-income countries. Best Pract Res Clin Rheumatol 2014;28:377-393.
- Frank C, Novokowsky B, Zummer M. Tool for Developing and Evaluating Models of Care: Created by the Musculoskeletal Stakeholder Community. Toronto: Arthritis Alliance of Canada; 2012.

- Kadu MK, Stolee P. Facilitators and barriers of implementing the chronic care model in primary care: a systematic review. BMC Family Practice 2015;16.
- World Health Organisation. Better non-communicable disease outcomes: Challenges and opportunities for health systems. Assessment Guide. Copenhagen: WHO Regional Office for Europe; 2014.
- **12.** Wagner EH, Austin BT, Davis C, Hindmarsh M, Schaefer J, Bonomi A. Improving chronic illness care: translating evidence into action. Health Aff 2001;20:64-78.
- **13.** Briggs AM, Jordan JE, Speerin R, et al. Models of care for musculoskeletal health: a cross-sectional qualitative study of Australian stakeholders' perspectives on relevance and standardised evaluation. BMC Health Serv Res 2015;15:509.
- 14. Graham ID, Logan J, Harrison MB, et al. Lost in knowledge translation: time for a map? J Contin Educ Health Prof 2006;26:13-24.
- **15.** Redman S, Turner T, Davies H, et al. The SPIRIT Action Framework: A structured approach to selecting and testing strategies to increase the use of research in policy. Social Sci Med 2015;136:147-55.
- 16. Brennan SE, Cumpston M, Misso ML, McDonald S, Murphy MJ, Green SE. Design and formative evaluation of the Policy Liaison Initiative: a long-term knowledge translation strategy to encourage and support the use of Cochrane systematic reviews for informing health policy. Evidence & Policy 2016;12:25-52.
- **17.** Lavis JN, Boyko JA, Gauvin FP. Evaluating deliberative dialogues focussed on healthy public policy. BMC Public Health 2014;14:1287.
- Moat KA, Lavis JN, Clancy SJ, El-Jardalid F, Pantojae T. Evidence briefs and deliberative dialogues: perceptions and intentions to act on what was learnt. Bull World Health Organ 2014;92:20-28.
- Michie S, van Stralen MM, West R. The behaviour change wheel: A new method for characterising and designing behaviour change interventions. Implementation Sci 2011;6.

- Stoffer MA, Smolen JS, Woolf A, et al. Development of patient-centred standards of care for osteoarthritis in Europe: the eumusc.net-project. Ann Rheum Dis 2015;74:1145-1149.
- **21.** Akesson K, Marsh D, Mitchell PJ, et al. Capture the Fracture: a Best Practice Framework and global campaign to break the fragility fracture cycle. Osteoporos Int 2013;24:2135-2152.
- Ganda K, Puech M, Chen JS, et al. Models of care for the secondary prevention of osteoporotic fractures: a systematic review and meta-analysis. Osteoporos Int 2013;24:393-406.
- Briggs AM, Sun W, Miller LJ, Geelhoed E, Huska A, Inderjeeth CA. Hospitalisations, admission costs and re-fracture risk related to osteoporosis in Western Australia are substantial: a 10-year review. Aust N Z J Public Health 2015;39:557-562.
- 24. Tulk C, Lane P, Gilbey A, et al. Improving osteoporosis management following minimal trauma fracture in a regional setting: The Coffs Fracture Card Project. Aust J Rural Health 2013;21:343-349.
- 25. NSW Agency for Clinical Innovation. Understanding Program Evaluation. An ACI Framework. Sydney: NSW Agency for Clinical Innovation; 2013.

Images have been sourced from: Page i (right), 6, 71: © Linda S. Nye Page i (left): © Shutterstock, graphixmania Page 13: © Shutterstock, MR.LIGHTMAN1975 Pages 19, 20, 31, 36, 50: © Shutterstock, sumkinn Page 23: © iStock, Razvan Page 49: © iStock, BraunS Page 64: © iStock, OJO_Images

Design: Nikki M Group www.nikkimgroup.com.au



